Author's response to reviews

Title: Evaluation of pulse-oximetry oxygen saturation taken through skin protective covering

Authors:

Jyotsna James (jyotsnajames@yahoo.com)
Lokesh Tiwari (lokeshdoc@yahoo.com)
Pramod K Upadhyay (pkumar@nii.res.in)
Vishnubhatla Sreenivas (sreenivas_vishnu@yahoo.com)
Vikas Bhambhani (vikasbhambhani@hotmail.com)
Jacob Puliyel (puliyel@vsnl.com)

Version: 4 Date: 30 January 2006

Author's response to reviews:

To
The Editor

We thank you for your letter and clarification. I am copying your letter below with the action taken in CAPITAL LETTERS

Thank you for all your help
Sincerely
Jacob Puliyel

Thank you for your email. Just to clarify, we are not concerned about the lack of ethical approval, however, please can you address Dr Dullenkopf's concerns by deleting the part about in-vitro testing from the manuscript?

DONE

This reviewer also notes that there are typing errors and English corrections that need to be made. Perhaps a colleague, preferably a native English speaker, could look it over for you.

DONE

Alternatively, here are the contact details of two professional copy editing services
http://www.abdn.ac.uk/mps

http://www.scientificmanager.com